Author's response to reviews

Title: Ultra-brief intervention for problem drinkers: Research Protocol

Authors:

John A Cunningham (john_cunningham@camh.net)
Clayton Neighbors (claytonn@u.washington.edu)
Cameron Wild (cam.wild@ualberta.ca)
Keith Humphreys (knh@stanford.edu)

Version: 2 Date: 17 August 2008

Author's response to reviews:

As requested, I have made the following revisions to the manuscript:

1) All references cited in the text and numbered in the order that they appear.
2) Author list same as they are in the submission system.
3) Email addresses added in requested format.
4) Removed: Keywords, Word-counts, Running head and Running title.
5) Underlining removed.
6) Abbreviations formatted as a paragraph.
7) Additional files mentioned in text in requested format.

Final check of manuscript also undertaken.

Thanks,

John